Author’s response to reviews

Title: Clinical outcomes of over-the-scope-clip system for the treatment of acute upper non-variceal gastrointestinal bleeding: a systematic review and meta-analysis

Authors:
Chunyu Zhong (447166418@qq.com)
Shali Tan (957242961@qq.com)
Yutang Ren (pumcren@126.com)
MuHan Lü (52721759@qq.com)
Yan Peng (1806857826@qq.com)
Xiangsheng Fu (drfuxs@gmail.com)
Xiaowei Tang (solitude5834@hotmail.com)

Version: 2 Date: 06 Dec 2019

Author’s response to reviews:

Dear Editors,

Thank you very much for considering publication of our manuscript. The reviewers have made great comments on our manuscript. After discussion with all the co-authors, we revised this manuscript seriously and carefully according to reviewers’ opinions. The answers to each reviewer are as follows:

Respond to Reviewer 2’ comments:
Kuang-Chun Hu, M.D. Ph.D. (Reviewer 2): After carefully review this paper, I think authors had done well in revised paper but had one mistake in figure 2: The arrow and proportion of acute upper non-variceal gastrointestinal bleeding etiology was unmatched and need corrected.
Answer to this question: Thanks for this comment. We are so sorry for this mistake in Figure 2. We already corrected the Figure 2.

In addition to correct this manuscript according to the reviewers’ advice, we also checked the manuscript time and time again to avoid grammatical or spelling errors throughout the article. Hope our corrected manuscript can be published in your journal soon. That will be our great honor.
Thank you very much! Best wishes!
yours sincerely,

Xiaowei Tang